Author's response to reviews

Title: Five years after the accident, whiplash casualties still have poorer quality of life in the physical domain than other mildly injured casualties: analysis of the ESPARR cohort.

Authors:

Charlène Tournier (tournier.charlene@wanadoo.fr)
Martine Hours (martine.hours@ifsttar.fr)
Pierrette Charnay (charnayp@gmail.com)
Laetitia Chossegros (chossegroslaetitia@yahoo.fr)
Hélène Tardy (helene.tardy@ifsttar.fr)

Version: 2 Date: 15 Dec 2015

Author’s response to reviews:

To the “BMC Public Health” journal

December 15th, 2015

Dear Sir or Madam,

We thank you for the potential acceptance of our paper entitled “Five years after the accident, whiplash casualties still have poorer quality of life in the physical domain than other mildly injured casualties: analysis of the ESPARR cohort” numbered PUBH-D-15-00721.

We carefully checked the editorial policies of your journal. To comply with it, we added a paragraph concerning the “Availability of Data “. We can’t publish our dataset, because we don’t have this possibility in the authorization given by the National Data Protection Authority (CNIL: CNIL Number 04-1417); but we may send a dataset to individual researchers who will send us a request by email.

We hope that this will comply with your rules.

Yours faithfully,

Charlène TOURNIER  Martine HOURS  Pierrette CHARNAY
Laetitia CHOSSEGROS  Hélène TARDY